Author’s response to reviews

Title: Experiences and needs of carers of Aboriginal children with a disability: a qualitative study

Authors:
Michelle DiGiacomo (Michelle.DiGiacomo@uts.edu.au)
Anna Green (anna.green-1@uts.edu.au)
Patricia Delaney (Patricia.Delaney@uts.edu.au)
John Delaney (John.Delaney@uts.edu.au)
Patrick Patradoon-Ho (Patrick.Patradoon-Ho@health.nsw.gov.au)
Patricia Davidson (pdavidson@jhu.edu)
Penelope Abbott (p.abbott@westernsydney.edu.au)

Version: 2 Date: 23 Oct 2017

Author’s response to reviews:

On behalf of all of the authors, thank you for the thoughtful and constructive comment. Please find below our response.

Editor comment: I am not sure what they authors mean by ‘Phenomenology-informed thematic analysis’. Please can they clarify what is meant by this in the paper or change the text so it states that they conducted a thematic analysis, which based on the description is what they did.

Response: We have changed the text so that it states thematic analysis was conducted (Abstract, p. 2; Analysis, p. 7).